**Title Page** 

# The role of gender pharmacogenetics in the personalization of drug treatment

Authors: Erika Cecchin, Bianca Posocco, Silvia Mezzalira, Marialuisa Appetecchia, Giuseppe Toffoli

Experimental and Clinical Pharmacology, Centro di Riferimento Oncologico di Aviano (CRO)

IRCCS, via Franco Gallini n. 2, 33081 Aviano PN, Italy (E.C., B.P., S.M., G.T.)

Oncological Endocrinology Unit, IRCCS Regina Elena National Cancer Institute - IFO, Rome,

Italy (M.A.)

# **Running Title Page**

# Gender pharmacogenetics in precision medicine

Corresponding Author: Giuseppe Toffoli, MD

Experimental and Clinical Pharmacology, CRO - IRCCS,

via Franco Gallini n. 2, 33081 Aviano PN, Italy,

office:(+39)-0434-659612

gtoffoli@cro.it

Text pages: 28

Tables: 2

Figure: 1

References: 80

Abstract: 163 words

Introduction: 416 words

Discussion: 3585 words

# **List of Abbreviations**

5- FU: 5-fluorouracil
ABC: ATP-Binding Cassette
ABRACL: Actin-Binding Rho Activating C-terminal Like
ACE: Angiotensin-Converting Enzyme
ADME: Absorption, Distribution, Metabolism, and Excretion
ADR: Adverse Drug Reaction
CD-40LG: Cluster of Differentiation 40 Ligand

COMT: Catechol-O-Methyltransferase **CPIC: Clinical Pharmacogenetics Implementation Consortium** CPNDS: Canadian Pharmacogenomics Network for Drug Safety CYP3A4, CYP2D6, CYP1A, CYP2C19: Cytochrome P450 3A4, 2D6, 1A, 2C19 DPWG: Dutch Pharmacogenetics Working Group *DPYD*: Dihydropyrimidine Dehydrogenase (gene) DPD: Dihydropyrimidine Dehydrogenase (protein) FoxP3: Forkhead box P3 ERCC1: Excision Repair Cross-Complementing isoform 1 ERCC2: Excision Repair Cross-Complementing isoform 2 FHL3: Four and a Half LIM domains 3 **FPs:** Fluoropyrimidines IL-2: Interleukin 2 IL28B: Interleukin 28B LILRA5: Leukocyte Immunoglobulin Like Receptor A5 MC1R: Melanocortin 1 Receptor MTHFR: Methylenetetrahydrofolate Reductase NBPF14: Neuroblastoma Breakpoint Family member 14 OPRD1: Opioid Receptor Delta 1 OPRM1: Opioid Receptor Mu 1 PD: Pharmacodynamics PGx: Pharmacogenetics **PK:** Pharmacokinetics PPi: Proton Pump inhibitor RBPMS2: RNA binding protein, mRNA processing factor 2 SNP: Single Nucleotide Polymorphism TLR-7: Toll Like Receptor 7 TRL-8: Toll Like Receptor 8 TYMS: Thymidylate Synthetase

Section Assignment: Metabolism, Transport, and Pharmacogenomics

## Abstract

The use of pharmacogenetic guidelines in personalizing treatments has shown the potential to reduce interindividual variability in drug response by enabling genotype-matched dosing and drug selection. However, other important factors, such as patient gender, may interact strongly with pharmacogenetics in determining the individual profile of toxicity and efficacy, but are still rarely considered when planning pharmacological treatment. The literature indicates that males and females respond differently to drugs, with women being at higher risk for toxicity and having different plasma exposure to drugs at standard doses. Recent studies have shown that pharmacogenetic variants may have different predictive value in different sexes, as in the case of treatment with opioids, ACE inhibitors, or proton pump inhibitors. Of particular interest is the case of treatment with fluoropyrimidines for cancer. A significant increase in toxicity has been described in female patients, with a more pronounced effect of specific *DPYD* and *TYMS* polymorphisms also noted. This manuscript reviews the major findings in the field of sex-specific pharmacogenomics.

## **Significance Statement**

Interindividual variability in drug response is an emerging issue in pharmacology. The genetic profile of patients, as well as their gender, may play a role in the identification of patients more exposed to the risk of adverse drug reactions or poor efficacy. This article reviews the current state of research on the interaction between gender and pharmacogenetics in addressing interindividual variability.

### Introduction

It is well known that there are significant interindividual differences in drug response and toxicity in patients treated with the same therapy. The safety and efficacy of drugs depend on many factors that influence their utilization in the body (Lichtman *et al.*, 2003; Wilkinson, 2005; Scripture and Figg, 2006). Drug-related interindividual variability factors include dosage, dosage form, and therapeutic regimen, which are responsible for drug-drug interactions. Patient-specific factors include age, gender, diet, natural physiological cycles, pregnancy, acute illness, liver and kidney dysfunction, and other chronic diseases (Huang and Temple, 2008). One of the principal causes of variability in drug effects is the patient's own metabolism. Approximately one third of variability among individuals in drug response is correlated with genetic single nucleotide polymorphisms (SNPs) in genes encoding phase 1 and phase 2 enzymes.

In the lengthy and costly development of drugs, patients' gender and genetics have been inadequately considered, resulting in biased dosing strategies for most of the available drugs. The lack of consideration of the genetic makeup of participating patients, which may lead to enrolment of patients in the clinical trial who are carriers of high-impact mutations, is emerging as a critical problem. In addition, analysis of clinical trial quality by several authors has shown that women are significantly underrepresented in pivotal phase 1, 2, and 3 clinical trials. Traditionally, studies have primarily included adult white, Caucasian males, particularly in phase I clinical trials (Fisher and Kalbaugh, 2011). The inclusion of women is central to all phases of drug development to rationalize the costs of the research and to improve drug safety and efficacy based on a personalized approach (Woodruff, 2014). Optimization of drugs in early phases of preclinical development, performed specifically on cells or mice of one sex, can lead to sex bias that is perpetuated in later phases of drug development. To get an idea of how topical this issue is, one might note, for example, that also for research on SARS-CoV-2 (COVID -19) at the moment only one of the four human cell lines used is female (Takayama, 2020).

In this context, personalized medicine or precision medicine is a model of healthcare that tailor pharmacological treatments and, more generally, medical interventions, to each individual based on the risk factors for treatment failure or for the disease (Mathur and Sutton, 2017; Manson *et al.*, 2017).

#### Pharmacogenetics and its implementation in the clinical practice

Personalized medicine has gained importance in recent years in light of increasing diagnostic and informatics approaches, particularly in the genetic field. In particular, the use of genetic information has played such a critical role in personalized medicine that the term pharmacogenomics was first used in the context of genetics and then evolved into a new branch of pharmacology (Burke *et al.*, 2014).

Pharmacogenomics and pharmacogenetics (PGx) aim to personalize therapy based on the genetic predisposition of patients (Pirmohamed, 2001; Katara and Yadav, 2019). According to many authors, adjusting dosages and medications to take genetic variants into account is the first true clinical application of genetics in the post-genomic era (Swen *et al.*, 2007).

Indeed, the presence of a variant in a gene encoding metabolic enzymes may increase or decrease its enzymatic activity, with significant implications for efficacy and/or the development of toxicity. Genetic variants could be related to i) lower or higher drug exposure, ii) increased toxic metabolite concentration, iii) altered interaction with the target drug, and iv) idiosyncratic drug toxicity due to immune activation (Pinto and Dolan, 2011). An estimated 90% of the population carries at least one variant in a gene related to drug metabolism or mechanism of action, or a variant indicating a higher risk for hypersensitivity reactions (Uetrecht, 2007; Van Driest *et al.*, 2014). The protein phenotype predicted based on the genetic profile of the encoding gene could be classified into the following groups: - "Ultrarapid Metabolizers" with increased enzyme activity; - "Normal Metabolizers" with a normal and weak enzyme activity; - "Intermediate Metabolizers" with an intermediate enzyme functionality; - "Poor Metabolizers", when the enzyme is completely dysfunctional (Caudle *et al.*, 2017).

It was reported that 7% of the 1200 drugs approved by the Food and Drug Administration (FDA) are linked to a clinically effective variant, and these drugs account for 18% of about 4 billion prescriptions in the United States (Budnitz *et al.*, 2006; Relling and Evans, 2015; Caraballo *et al.*,

2017). Consequently, the inability to obtain appropriate therapy, even when PGx information is used, means the so-called "therapeutic odyssey" for the patient. This condition is characterized by the constant search for the most effective therapy, ineffective treatments, frequent visits to the doctor, alternative treatment regimens based on the use of dietary supplements, phytotherapeutics, or polytherapeutic agents, the deterioration of the patient's condition, and the failure of treatment. PGx testing before treatment could curb this phenomenon and many of the above cases could be identified a priori to avoid side effects or therapeutic inefficacy (Lazaridis, 2017).

Currently, the slow clinical adoption of PGx is primarily due to barriers that delay its introduction into clinical practice. The lack of standardized PGx guidelines and their difficult interpretation by clinicians who have been resistant to PGx information are just some of these obstacles (Swen et al., 2007; Dunnenberger et al., 2015). Nowadays, authoritative consortia worldwide have developed guidelines that can help clinicians translate available genetic test results into clinical decisions. The Royal Dutch Association for the Advancement of Pharmacy - Pharmacogenetics Working Group (DPWG), Clinical Pharmacogenetics Implementation Consortium (CPIC), the Canadian Pharmacogenomics Network for Drug Safety (CPNDS), and additional scientific consortia are working to facilitate the incorporation of PGx testing into clinical practice and to translate genetic information into prescribing recommendations (Clinical Pharmacogenetics Implementation Consortium, https://cpicpgx.org) (Swen et al., 2008, 2018; Caudle et al., 2014; Abdullah-Koolmees et al., 2020). The PharmGKB website, a National Institutes of Health (NIH)-funded resource, collects international guidelines and provides information on how genetic variations affect drug response, making this knowledge accessible to clinicians and researchers. To date, 249 clinical guidelines by either CPIC, DPWG, or other consortia are available for 147 drugs (https://www.pharmgkb.org).

Since the early 2000s, PGx clinical programs with implementation models have been reported in various contexts of healthcare, using both preventive testing or at the point-of-care (Luczak *et al.*, 2021). In this context, the European Union-funded Ubiquitous Pharmacogenomics (U-Pgx) study

tested the impact of implementing PGx guidelines on the safety of a list of 43 commonly prescribed drugs at 7 clinical sites in Europe as part of a randomized and prospective clinical trial (PREemptive Pharmacogenomic testing for prevention of Adverse drug Reactions - PREPARE) (Cecchin *et al.*, 2017; Swen *et al.*, 2023).

#### Sex medicine in the context of personalized medicine

Among all drug classes, women are almost 2-fold more likely than men to experience adverse drug reactions (ADRs) and at increased risk of hospitalization for an ADR (Zucker and Prendergast, 2020; Madla *et al.*, 2021; Zucker *et al.*, 2022). Many drugs administered to both sexes at the same dose should be re-evaluated for gender-specific dose adjustment, such as the representative case of the sedative-hypnotic zolpidem, for which gender-specific dose adjustment was required after decades in which post marketing reports highlighted that women were reporting cognitive deficits when receiving the standard dose of drug for male (Zucker *et al.*, 2022). In the United States, between 1997 and 2000, among eight of 10 drugs were withdrawn from the market because of very severe side effects encountered mostly in women (Schiebinger *et al.*, 2016). Sex medicine is an innovative approach to precision medicine that aims to improve care and treatment for both men and women, overcoming milligram/kilogram-based or "one size fits all" drug administration, which often leads to treatment errors (Baggio *et al.*, 2013; Wagner *et al.*, 2019). Sex differences affecting virtually all body areas correlate with protection from or susceptibility to cancer, cancer progression, and response to therapy (Dong *et al.*, 2020; Zucker and Prendergast, 2020).

Patient sex could influence both PK and PD drug pathways, leading to sex differences in ADRs (Soldin and Mattison, 2009; Soldin *et al.*, 2011; Dong *et al.*, 2020). For most of the FDA-approved drugs studied (88%) for which sex-specific data were available PK, increased blood concentrations and longer elimination times were found in women in more than 10 therapeutic categories, and these PKs were strongly associated with sex differences in ADRs. FDA reviewed 300 new drug

#### JPET Fast Forward. Published on March 31, 2023 as DOI: 10.1124/jpet.122.001416 This article has not been copyedited and formatted. The final version may differ from this version.

applications between 1995 and 2000, and only 54% included an analysis of gender. In addition, 8% of these drugs presented a difference of around 40% in pharmacokinetics when comparing males and females, yet gender was not accounted for in dosing recommendations (Anderson, 2005). The lack of published sex-specific data regarding PK for many different drugs raises concerns that other drugs could present PK differences according to sex but data are not made publicly available (Zucker and Prendergast, 2020; Zucker *et al.*, 2022).

Women are more prone to overdose due to lower volume of distribution, higher body fat percentage (13.5-16.5 kg in women), larger free fraction of the drug, and slower xenobiotic clearance. Body composition parameters that could affect drug distribution include body fluids composition and distribution which volume is greater in males than females, resulting in different responses to the drug. Plasma protein binding has also been shown to vary between the two sexes due to the influence of estrogens, which increase the concentration of serum binding globulins. The use of oral contraceptives, pregnancy or menopause status are cited as sex-specific conditions that affect the PK and PD of various drugs. Another important difference between the two sexes is related to the composition of the gut microbiota (Mueller *et al.*, 2006; Wen *et al.*, 2008). In addition, lifestyle factors such as use of tobacco or excessive alcohol consumption, diet and physical inactivity, which are known to have a direct impact on drug response, differ greatly between men and women, as reported in several studies. In addition, women tend to take a greater number of medications than men, leading to increased drug-drug interactions.

Sex hormones have secondary effects that up- or down-regulate genes contributing to exacerbate gender differences in treatment outcomes (Pinsonneault and Sadée, 2003). Transcriptional processes promoting the expression of absorption, distribution, metabolism, and excretion (ADME) genes may be activated by estrogens by binding specific nuclear receptors causing a significant difference in the activity of phase I and phase II metabolic enzymes by sex. The phase I enzymes CYP3A4 and CYP2D6 are more active in females, while CYP1A family enzymes have higher metabolic activity

in males. These differences result in sex-specific changes in exposure to various drugs, including clomipramine, clozapine, olanzapine, acetaminophen, codeine, diazepam, fentanyl, statins, and tamoxifen. Phase 2 enzymes such as UDP-glucuronosyltransferases and methyltransferases have been found to be more active in males than females and directly affect the clearance of ibuprofen, acetaminophen, azathioprine, dopamine, oxazepam, and levodopa (Yang *et al.*, 2012; Franconi and Campesi, 2014).

## The role of sex medicine in the pharmacogenomics era

In the context of the clearly different ability of men and women to metabolise drugs, the role of PGx may become crucial. Indeed, a metabolic background already altered by sex-specific factors could enhance the effect of specific genetic polymorphisms or otherwise render them negligible (Figure 1). Many genetic polymorphisms have been shown to exhibit a sex-specific effect (Myburgh *et al.*, 2012). A recent genome-wide association study (GWAS) focused on identifying predictive markers associated with psychiatric disorders identified genetic variants that are differentially predictive in men and women. Variants in genes related to neuronal development, the immune system, and vascular function have been associated with sex-dependent effects on the onset of schizophrenia, major depressive disorder, and bipolar disorder (Blokland *et al.*, 2022). However, there are no precise studies on the gender effects of PGx on patient treatment, as in other areas of pharmacological research.

The most striking difference between males and females is sex chromosomes. Indeed, an impaired drug response was previously related to large genetic aberrations in Y chromosome, inactivation of X chromosomes and other genetic and epigenetic alterations (Carè *et al.*, 2018). Immune function is regulated by genes located in the X chromosome (e.g., *TLR*, cytokine receptors, transcription factors), therefore immune genes can be up-regulated in females due to X chromosome inactivation, which may explain the gender discrepancy in immunotherapy. Specifically, IL -2, TLR

-7, TRL-8, CD-40LG, and FoxP3 are X genes related to the immune response which can lead to higher resistance to immunotherapy in females due to X escape, (Irelli *et al.*, 2020). Differential response to infection between males and females is one of the aspects related to sex differences in the immune system activation. Response to HCV therapy was related to four genetic polymorphisms near the *IL28B* gene in a sex-dependent manner. Women had a better response to treatment and HCV clearance than men (Rao *et al.*, 2012).

Exposure to many drugs was affected by patients sex and related enzymatic activities in several cases such as irinotecan, ibuprofen, acetaminophen, dopamine, oxazepam and azathioprine metabolised by UGTs; clomipramine, acetaminophen and clozapine, metabolized by CYP1A; diazepam, lovastatin, simvastatin and fentanyl metabolized by CYP3A; fluoxetine, codeine, tamoxifen, encainide and flecainide, metabolized by CYP2D6 (Soldin and Mattison, 2009).

The volume of distribution of drugs could be sex-dependently affected by polymorphisms in the ABC genes as well as their glomerular filtration (Anderson, 2008).

Many genetic polymorphisms have been associated with sex differences in drug response (Table 1). An interaction between sex and genetics of patients has been highlighted for *MC1R* gene in analgesic therapy. When pentazocine was administered to fair-skinned, red-haired women in comparison with red-haired men and women who did not have the variant alleles, better analgesia was associated with the presence of two variant alleles of the *MC1R* in their genome (Mogil *et al.*, 2003). In a study conducted in 582 opioid-dependent patients randomized to receive methadone or buprenorphine/naloxone, no overall effect of genetic polymorphisms in *OPRD1* was observed, but a specific sex-specific effect was noted for 2 intronic SNPs (rs581111 and rs529520) that were related to the outcome of the treatment only in women treated with buprenorphine (Clarke et al., 2014). In addition, a genotype analysis of patients treated with opioids, antipsychotics and antiepileptic drugs for noncancer chronic pain, showed that the nature of adverse drug reactions (ADRs) differed between males and females according to the patients genotype: *OPRM1*-G allele and *COMT*-GG

genotype were associated with specific side effects (in men vomiting and depression, whereas in women dizziness and dry skin). The sexual dysfunction incidence was the same in both sexes (Planelles *et al.*, 2020). A recent observational study compared the use of tapentadol (n=194) or oxycodone/naloxone (n=175) with the prescription of other opioids for the treatment of chronic noncancer pain in 585 real-world patients. In addition to clinical endpoints of treatment outcomes, some genetic variants of the OPRM1 (rs1799971, A118G) and COMT (rs4680, G472A) genes were analyzed. In patients treated with oxycodone/naloxone, patients with the homozygous genotype COMT-472AA had a higher rate of erythema and vomiting, especially in women, again indicating a gender effect (Barrachina et al., 2022). Transcriptomics signatures was analysed in male and female bipolar patients. Although the study did not focus on germline genetic polymorphisms, the authors showed that two genes (*RBPMS2* and *LILRA5*) were selectively expressed in men that responded to lithium treatment whereas three genes (ABRACL, FHL3, and NBPF14) were expressed in female lithium responders (Eugene et al., 2018). About the treatment of cardiovascular disease, polymorphisms in ACE gene were associated with a sex-specific differential outcome after treatment with ACE inhibitors (ACEIs). They are more effective in women with genotype D/D compared with men with genotype D/D and more effective in men with genotype D/D than in men with genotype I/D or I/I (Ruggenenti et al., 2000). Hydrochlorothiazide response was also affected by ACE polymorphism in a sex-specific manner and I/I women responded better than D/D men (Schwartz *et al.*, 2002). Male patients carrying an allelic variant of ABCC2-24C > T and treated with atorvastatin for hypercholesterolemia, had a significant lower response than in women carrying the same genetic variant (Prado et al., 2018).

More recently, *CYP2C19*, involved in the clearance of proton pump inhibitors (PPIs) (such as pantoprazole, rabeprazole, omeprazole, lansoprazole and esomeprazole) was studied in relation to the prevalence of migraine in the large UK Biobank cohort. A higher migraine prevalence at baseline (odds ratio [OR] = 1.25, p < 0.0001) was associated to the *CYP2C19* metabolizer

phenotype in patients treated with PPIs. In particular, a poor *CYP2C19* metabolizer status was associated with chronic migraine only in men, suggesting a potentially relevant role of the interaction between gender and *CYP2C19* phenotype (Pisanu *et al.*, 2021).

#### The effect of pharmacogenetics- sex interaction on cancer treatment with fluoropyrimidines

Fluoropyrimidines (FPs), including 5-fluorouracil (5- FU) and capecitabine, are widely used in the treatment of tumors and represent an important example of gender disparity in treatment outcomes (Stein et al., 1995; Sloan et al., 2000; Mueller et al., 2013). Clearance of FPs is lower in women than in men (Milano et al., 1992), resulting in higher toxicity in women and compromising therapeutic efficacy (Mader, 2007; Wang and Huang, 2007). Zalcberg et al. reported as early as 1998 that hematologic and nonhematologic toxicities were influenced by gender after administration of 5 FU and leucovorin in patients with advanced colorectal cancer (Zalcberg et al., 1998). Severe toxicities such as stomatitis, leukopenia, alopecia, and diarrhea occurred more frequently in women compared with men during 5- FU based chemotherapy. These data were recently confirmed by Wagner et al. in an analysis of women with colon cancer who received adjuvant FP -based chemotherapy (Wagner et al., 2021). The mechanism underlying the genderdependent PK of 5- FU when administered as an infusion, with lower drug clearance in women, is still unclear, but the authors postulated a role in DPD activity (Milano et al., 1992). DPD activity has been shown to be decreased by 15% in women (Etienne et al., 1994). This may pave the way to investigate the role of gender in predicting *DPYD* variants in FP-related toxicity, as the proportion of toxicity cases explained by DPYD variants may be different in men and women (Amstutz et al., 2011). Lower DPD activity associated with toxicity in females could be explained by DPD enzymatic deficiency (Milano et al., 1999). Four DPYD variants (DPYD\*2A, DPYD\*13, DPYD c.2846C > T and DPYD-HapB3) have been shown to functionally affect DPD functionality and increase the risk of severe toxicity associated with FPs administration. All international guidelines now agree that pretreatment *DPYD* genotyping is recommended (Amstutz *et al.*, 2018; Lunenburg *et al.*, 2020), and recently the EMA also published a document recommending DPD screening to increase treatment safety (https://www.ema.europa.eu/en/news/ema-recommendationsdpd-testing-prior-treatment-fluorouracil-capecitabine-tegafur-flucytosine). According to current guidelines, patients who are carriers of an allelic variant within the panel should receive a halved initial dose of FPs. If there is a remote possibility that a patient is a carrier of more than one variant allele, patients should be phenotyped (i.e., DPD activity should be determined); otherwise, treatment with FPs should be avoided.

A potential interaction between *DPYD* polymorphisms and patients sex has been investigated (Table 2). In a study by Schwab et al. it was shown that the effect of the *DPYD\*2A* variant on toxicity associated with 5 FU can be considered gender-specific (Schwab *et al.*, 2008). While a significant association with this variant was observed in male patients, no effect was seen in female patients. Similarly, Lee et al. observed a stronger effect of *DPYD\*2A* in males compared to females, suggesting an interaction between gender and *DPYD* polymorphisms in the context of treatment with FPs (Lee *et al.*, 2014).

Other genes and the sex-specific effect of their polymorphisms have been investigated for FPsbased treatments (Table 2). In a multicenter, randomised, noninferiority phase III study conducted in patients with high-risk stage II / III colon cancer treated with 6 vs. 3 months of adjuvant chemotherapy with FOLFOX-4 or XELOX (TOSCA), Ruzzo A.M. and colleagues analysed how the interaction between 17 genetic polymorphisms and patient gender might affect treatment-related toxicity. An interaction of rs1801133 (*MTHFR*) and rs1799793 (*ERCC2*) on time to onset of grade 3 hematologic toxicity, of rs13181 (*ERCC2*) on time to onset of grade 3 gastrointestinal toxicity, and of rs11615 (*ERCC1*) on time to onset of grade 3 neurologic toxicity was found. The rs1799793 genotype GA (p=0.006) and the A allele (p=0.009) specifically decreased the time to onset of grade 3 hematologic toxicities in males, while the rs11615 genotype CC worsened the time to onset of grade 3 neurologic toxicities and the rs13181 G allele improved the time to onset of grade 3 gastrointestinal toxicities in females (Ruzzo *et al.*, 2019). In this case, the effect of genes in DNA repair pathways may reflect the effect of oxaliplatin in the combination treatment that patients received.

On the other hand, the sex-specific effect on FPs toxicity observed for the *MTHFR* polymorphism was recently confirmed by Ioannou et al. who genotyped 313 cancer patients treated with FPs for the same *MTHFR* polymorphism rs1801133. They found a specific effect of the polymorphism in female patients, in whom the *MTHFR* c.665 CT and TT genotypes were associated with both the need for dose reduction due to toxicity and the percentage of dose reduction. Such differences were not present in male patients, again confirming that genetic polymorphisms have a stronger influence on toxicity risk in females instead of males (Ioannou *et al.*, 2022).

Although the role of the *TYMS* polymorphism in predicting severe adverse effects in patients treated with FPs is still debated, a higher rate of ADRs in patients with *TYMS*-TSER 3R/2R genotype and treated with 5 FU /capecitabine was observed in female cancer patients compared to males, possibly related to ER regulation of TS expression (Ioannou *et al.*, 2021). *TYMS* genotype 2R/2R was found to have a higher prevalence in African American female patients with adverse effects compared to male patients (Khushman *et al.*, 2021).

#### **Conclusion and future perspectives**

Pgx is becoming increasingly important in clinics, and several consortia have made specific recommendations for genotype-based prescribing. In this context, the impact of patient sex may be important, as it is reported to affect the clinical outcome of pharmacological treatment, in terms of both toxicity and efficacy. This is often mediated by a different ADME of drugs in men and women-a 40% difference between the two sexes (PK) -and by a different reactive immune system. However, sex-specific PGx recommendations that take into account both aspects have not yet been formulated. Evidence shows that women have a doubled risk to develop an ADR compared to men with higher hospitalization rates (Tharpe, 2011). There is growing evidence that the effect of genetic polymorphisms may differ in different gender contexts. There are limited and anecdotal data on the effects of gender on PGx-directed drug therapy derived from clinical investigations. These limitations could be overcome, ensuring adequate representation of both genders in clinical trials. Regulatory agencies regularly conduct analyzes to verify gender balance in recruitment, but in general the primary and secondary endpoints do not consider separate gender analyzes, particularly when addressing gender differences in PGx.

In recent years, the impact of germline genetic variants on therapeutic outcomes has been increasingly studied and is beginning to be used in clinical practice to improve drug efficacy and safety. Although there is evidence of sex differences in the efficacy and toxicity of many drugs, including chemotherapeutics, sex medicine still remains a mirage. Personalized and precision medicine should in the future assess how PGx variants might have a different role in men and women.

**Figure 1.** The different background of males and females in terms of drug pharmacokinetics and pharmacodynamics interacts with the genetics of specific enzyme players, such as cytochromes and intracellular transporters, causing sex-specific pharmacogenomic associations. There is a need to develop sex-based pharmacogenomics guidelines to address this topic.

# **Authorship Contributions**

Participated in research design: Cecchin, Toffoli

Performed data analysis: Mezzalira, Posocco

Wrote or contributed to the writing of the manuscript: Cecchin, Posocco, Toffoli, Appetecchia

## References

- Abdullah-Koolmees H, van Keulen AM, Nijenhuis M, and Deneer VHM (2020) Pharmacogenetics Guidelines: Overview and Comparison of the DPWG, CPIC, CPNDS, and RNPGx Guidelines. *Front Pharmacol* **11**:595219.
- Amstutz U, Froehlich TK, and Largiadèr CR (2011) Dihydropyrimidine dehydrogenase gene as a major predictor of severe 5-fluorouracil toxicity. *Pharmacogenomics* **12**:1321–1336.
- Amstutz U, Henricks LM, Offer SM, Barbarino J, Schellens JHM, Swen JJ, Klein TE, McLeod HL, Caudle KE, Diasio RB, and Schwab M (2018) Clinical Pharmacogenetics Implementation Consortium (CPIC) Guideline for Dihydropyrimidine Dehydrogenase Genotype and Fluoropyrimidine Dosing: 2017 Update. *Clin Pharmacol Ther* **103**:210–216.
- Anderson GD (2008) Gender differences in pharmacological response. Int Rev Neurobiol 83:1-10.
- Anderson GD (2005) Sex and racial differences in pharmacological response: where is the evidence? Pharmacogenetics, pharmacokinetics, and pharmacodynamics. *J Womens Health (Larchmt)* **14**:19–29.
- Baggio G, Corsini A, Floreani A, Giannini S, and Zagonel V (2013) Gender medicine: a task for the third millennium. *Clin Chem Lab Med* **51**:713–727.
- Barrachina J, Margarit C, Muriel J, López-Gil S, López-Gil V, Vara-González A, Planelles B, Inda M-D-M, Morales D, and Peiró AM (2022) Oxycodone/naloxone versus tapentadol in real-world chronic noncancer pain management: an observational and pharmacogenetic study. *Sci Rep* **12**:10126.
- Blokland GAM, Grove J, Chen C-Y, Cotsapas C, Tobet S, Handa R, Schizophrenia Working Group of the Psychiatric Genomics Consortium, St Clair D, Lencz T, Mowry BJ, Periyasamy S, Cairns MJ, Tooney PA, Wu JQ, Kelly B, Kirov G, Sullivan PF, Corvin A, Riley BP, Esko T, Milani L, Jönsson EG, Palotie A, Ehrenreich H, Begemann M, Steixner-Kumar A, Sham PC, Iwata N, Weinberger DR, Gejman PV, Sanders AR, Buxbaum JD, Rujescu D, Giegling I, Konte B, Hartmann AM, Bramon E, Murray RM, Pato MT, Lee J, Melle I, Molden E, Ophoff RA, McQuillin A, Bass NJ, Adolfsson R, Malhotra AK, Bipolar Disorder Working Group of the Psychiatric Genomics Consortium, Martin NG, Fullerton JM, Mitchell PB, Schofield PR, Forstner AJ, Degenhardt F, Schaupp S, Comes AL, Kogevinas M, Guzman-Parra J, Reif A, Streit F, Sirignano L, Cichon S, Grigoroiu-Serbanescu M, Hauser J, Lissowska J, Mayoral F, Müller-Myhsok B, Świątkowska B, Schulze TG, Nöthen MM, Rietschel M, Kelsoe J, Leboyer M, Jamain S, Etain B, Bellivier F, Vincent JB, Alda M, O'Donovan C, Cervantes P, Biernacka JM, Frye M, McElroy SL, Scott LJ, Stahl EA, Landén M, Hamshere ML, Smeland OB, Djurovic S, Vaaler AE, Andreassen OA, Major Depressive Disorder Working Group of the Psychiatric Genomics Consortium, Baune BT, Air T, Preisig M, Uher R, Levinson DF, Weissman MM, et al. (2022) Sex-Dependent Shared and Nonshared Genetic Architecture Across Mood and Psychotic Disorders. Biol *Psychiatry* **91**:102–117.
- Budnitz DS, Pollock DA, Weidenbach KN, Mendelsohn AB, Schroeder TJ, and Annest JL (2006) National surveillance of emergency department visits for outpatient adverse drug events. *JAMA* **296**:1858–1866.
- Burke W, Brown Trinidad S, and Press NA (2014) Essential elements of personalized medicine. Urol Oncol **32**:193–197.

- Caraballo PJ, Bielinski SJ, St Sauver JL, and Weinshilboum RM (2017) Electronic Medical Record-Integrated Pharmacogenomics and Related Clinical Decision Support Concepts. *Clin Pharmacol Ther* **102**:254–264.
- Carè A, Bellenghi M, Matarrese P, Gabriele L, Salvioli S, and Malorni W (2018) Sex disparity in cancer: roles of microRNAs and related functional players. *Cell Death Differ* **25**:477–485.
- Caudle KE, Dunnenberger HM, Freimuth RR, Peterson JF, Burlison JD, Whirl-Carrillo M, Scott SA, Rehm HL, Williams MS, Klein TE, Relling MV, and Hoffman JM (2017) Standardizing terms for clinical pharmacogenetic test results: consensus terms from the Clinical Pharmacogenetics Implementation Consortium (CPIC). *Genet Med* **19**:215–223.
- Caudle KE, Klein TE, Hoffman JM, Muller DJ, Whirl-Carrillo M, Gong L, McDonagh EM, Sangkuhl K, Thorn CF, Schwab M, Agundez JAG, Freimuth RR, Huser V, Lee MTM, Iwuchukwu OF, Crews KR, Scott SA, Wadelius M, Swen JJ, Tyndale RF, Stein CM, Roden D, Relling MV, Williams MS, and Johnson SG (2014) Incorporation of pharmacogenomics into routine clinical practice: the Clinical Pharmacogenetics Implementation Consortium (CPIC) guideline development process. *Curr Drug Metab* 15:209–217.
- Cecchin E, Roncato R, Guchelaar HJ, Toffoli G, and Ubiquitous Pharmacogenomics Consortium (2017) Ubiquitous Pharmacogenomics (U-PGx): The Time for Implementation is Now. An Horizon2020 Program to Drive Pharmacogenomics into Clinical Practice. *Curr Pharm Biotechnol* **18**:204–209.
- Clarke T-K, Crist RC, Ang A, Ambrose-Lanci LM, Lohoff FW, Saxon AJ, Ling W, Hillhouse MP, Bruce RD, Woody G, and Berrettini WH (2014) Genetic variation in OPRD1 and the response to treatment for opioid dependence with buprenorphine in European-American females. *Pharmacogenomics J* **14**:303–308.
- Dong M, Cioffi G, Wang J, Waite KA, Ostrom QT, Kruchko C, Lathia JD, Rubin JB, Berens ME, Connor J, and Barnholtz-Sloan JS (2020) Sex Differences in Cancer Incidence and Survival: A Pan-Cancer Analysis. *Cancer Epidemiol Biomarkers Prev* **29**:1389–1397.
- Dunnenberger HM, Crews KR, Hoffman JM, Caudle KE, Broeckel U, Howard SC, Hunkler RJ, Klein TE, Evans WE, and Relling MV (2015) Preemptive clinical pharmacogenetics implementation: current programs in five US medical centers. *Annu Rev Pharmacol Toxicol* **55**:89–106.
- Etienne MC, Lagrange JL, Dassonville O, Fleming R, Thyss A, Renée N, Schneider M, Demard F, and Milano G (1994) Population study of dihydropyrimidine dehydrogenase in cancer patients. J Clin Oncol 12:2248–2253.
- Eugene AR, Masiak J, and Eugene B (2018) Predicting lithium treatment response in bipolar patients using gender-specific gene expression biomarkers and machine learning. *F1000Res* **7**:474.
- Fisher JA, and Kalbaugh CA (2011) Challenging assumptions about minority participation in US clinical research. *Am J Public Health* **101**:2217–2222.
- Franconi F, and Campesi I (2014) Pharmacogenomics, pharmacokinetics and pharmacodynamics: interaction with biological differences between men and women. *Br J Pharmacol* **171**:580–594.
- Huang S-M, and Temple R (2008) Is this the drug or dose for you? Impact and consideration of ethnic factors in global drug development, regulatory review, and clinical practice. *Clin Pharmacol Ther* **84**:287–294.

- Ioannou C, Ragia G, Balgkouranidou I, Xenidis N, Amarantidis K, Koukaki T, Biziota E, Kakolyris S, and Manolopoulos VG (2021) Gender-dependent association of TYMS-TSER polymorphism with 5fluorouracil or capecitabine-based chemotherapy toxicity. *Pharmacogenomics* **22**:669–680.
- Ioannou C, Ragia G, Balgkouranidou I, Xenidis N, Amarantidis K, Koukaki T, Biziota E, Kakolyris S, and Manolopoulos VG (2022) MTHFR c.665C>T guided fluoropyrimidine therapy in cancer: genderdependent effect on dose requirements. Drug Metab Pers Ther 37:323–327.
- Irelli A, Sirufo MM, D'Ugo C, Ginaldi L, and De Martinis M (2020) Sex and Gender Influences on Cancer Immunotherapy Response. *Biomedicines* **8**:232.
- Katara P, and Yadav A (2019) Pharmacogenes (PGx-genes): Current understanding and future directions. *Gene* **718**:144050.
- Khushman M, Patel GK, Maharjan AS, McMillin GA, Nelson C, Hosein P, and Singh AP (2021) The prevalence and clinical relevance of 2R/2R TYMS genotype in patients with gastrointestinal malignancies treated with fluoropyrimidine-based chemotherapy regimens. *Pharmacogenomics J* **21**:308–317.
- Lazaridis KN (2017) Improving Therapeutic Odyssey: Preemptive Pharmacogenomics Utility in Patient Care. *Clin Pharmacol Ther* **101**:39–41.
- Lee AM, Shi Q, Pavey E, Alberts SR, Sargent DJ, Sinicrope FA, Berenberg JL, Goldberg RM, and Diasio RB (2014) DPYD Variants as Predictors of 5-fluorouracil Toxicity in Adjuvant Colon Cancer Treatment (NCCTG N0147). J Natl Cancer Inst **106**:dju298.
- Lichtman SM, Skirvin JA, and Vemulapalli S (2003) Pharmacology of antineoplastic agents in older cancer patients. *Crit Rev Oncol Hematol* **46**:101–114.
- Luczak T, Brown SJ, Armbruster D, Hundertmark M, Brown J, and Stenehjem D (2021) Strategies and settings of clinical pharmacogenetic implementation: a scoping review of pharmacogenetics programs. *Pharmacogenomics* **22**:345–364.
- Lunenburg CATC, van der Wouden CH, Nijenhuis M, Crommentuijn-van Rhenen MH, de Boer-Veger NJ, Buunk AM, Houwink EJF, Mulder H, Rongen GA, van Schaik RHN, van der Weide J, Wilffert B, Deneer VHM, Swen JJ, and Guchelaar H-J (2020) Dutch Pharmacogenetics Working Group (DPWG) guideline for the gene-drug interaction of DPYD and fluoropyrimidines. *Eur J Hum Genet* **28**:508– 517.
- Mader RM (2007) The potential of gender-specific tumor pharmacology. *Pharmacogenomics* 8:271–274, Future Medicine.
- Madla CM, Gavins FKH, Merchant HA, Orlu M, Murdan S, and Basit AW (2021) Let's talk about sex: Differences in drug therapy in males and females. *Advanced Drug Delivery Reviews* **175**:113804.
- Manson LE, van der Wouden CH, Swen JJ, and Guchelaar H-J (2017) The Ubiquitous Pharmacogenomics consortium: making effective treatment optimization accessible to every European citizen. *Pharmacogenomics* **18**:1041–1045, Future Medicine.
- Mathur S, and Sutton J (2017) Personalized medicine could transform healthcare. Biomed Rep 7:3–5.
- Milano G, Etienne MC, Cassuto-Viguier E, Thyss A, Santini J, Frenay M, Renee N, Schneider M, and Demard F (1992) Influence of sex and age on fluorouracil clearance. *J Clin Oncol* **10**:1171–1175.

- Milano G, Etienne MC, Pierrefite V, Barberi-Heyob M, Deporte-Fety R, and Renée N (1999) Dihydropyrimidine dehydrogenase deficiency and fluorouracil-related toxicity. *Br J Cancer* **79**:627–630.
- Mogil JS, Wilson SG, Chesler EJ, Rankin AL, Nemmani KVS, Lariviere WR, Groce MK, Wallace MR, Kaplan L, Staud R, Ness TJ, Glover TL, Stankova M, Mayorov A, Hruby VJ, Grisel JE, and Fillingim RB (2003) The melanocortin-1 receptor gene mediates female-specific mechanisms of analgesia in mice and humans. *Proc Natl Acad Sci U S A* **100**:4867–4872.
- Mueller F, Büchel B, Köberle D, Schürch S, Pfister B, Krähenbühl S, Froehlich TK, Largiader CR, and Joerger M (2013) Gender-specific elimination of continuous-infusional 5-fluorouracil in patients with gastrointestinal malignancies: results from a prospective population pharmacokinetic study. *Cancer Chemother Pharmacol* **71**:361–370.
- Mueller S, Saunier K, Hanisch C, Norin E, Alm L, Midtvedt T, Cresci A, Silvi S, Orpianesi C, Verdenelli MC, Clavel T, Koebnick C, Zunft H-JF, Doré J, and Blaut M (2006) Differences in fecal microbiota in different European study populations in relation to age, gender, and country: a cross-sectional study. *Appl Environ Microbiol* **72**:1027–1033.
- Myburgh R, Hochfeld WE, Dodgen TM, Ker J, and Pepper MS (2012) Cardiovascular pharmacogenetics. *Pharmacol Ther* **133**:280–290.
- Pinsonneault J, and Sadée W (2003) Pharmacogenomics of multigenic diseases: sex-specific differences in disease and treatment outcome. AAPS PharmSci 5:E29.
- Pinto N, and Dolan ME (2011) Clinically relevant genetic variations in drug metabolizing enzymes. *Curr Drug Metab* **12**:487–497.
- Pirmohamed M (2001) Pharmacogenetics and pharmacogenomics. Br J Clin Pharmacol 52:345–347.
- Pisanu C, Welander NZ, Rukh G, Schiöth HB, and Mwinyi J (2021) Association between migraine prevalence, treatment with proton-pump inhibitors and CYP2C19 phenotypes in UK Biobank. *Biomed Pharmacother* **143**:112234.
- Planelles B, Margarit C, Inda M-M, Ballester P, Muriel J, Barrachina J, Ajo R, Esteban M-D, and Peiró AM (2020) Gender based differences, pharmacogenetics and adverse events in chronic pain management. *Pharmacogenomics J* 20:320–328, Nature Publishing Group.
- Prado Y, Arencibia A, Zambrano T, and Salazar LA (2018) Gender-Specific Association Between ABCC2 -24C>T SNP and Reduction in Triglycerides in Chilean Patients Treated With Atorvastatin. *Basic Clin Pharmacol Toxicol* **122**:517–522.
- Rao H-Y, Sun D-G, Jiang D, Yang R-F, Guo F, Wang J-H, Liu F, Zhang H-Y, Zhang H-H, Du S-C, Jin Q, Qin H, Lok A-S-F, and Wei L (2012) IL28B genetic variants and gender are associated with spontaneous clearance of hepatitis C virus infection. J Viral Hepat 19:173–181.
- Relling MV, and Evans WE (2015) Pharmacogenomics in the clinic. Nature 526:343–350.
- Ruggenenti P, Perna A, Zoccali C, Gherardi G, Benini R, Testa A, and Remuzzi G (2000) Chronic proteinuric nephropathies. II. Outcomes and response to treatment in a prospective cohort of 352 patients: differences between women and men in relation to the ACE gene polymorphism. Gruppo Italiano di Studi Epidemologici in Nefrologia (Gisen). J Am Soc Nephrol **11**:88–96.

- Ruzzo A, Graziano F, Galli Francesca, Galli Fabio, Rulli E, Lonardi S, Ronzoni M, Massidda B, Zagonel V, Pella N, Mucciarini C, Labianca R, Ionta MT, Bagaloni I, Veltri E, Sozzi P, Barni S, Ricci V, Foltran L, Nicolini M, Biondi E, Bramati A, Turci D, Lazzarelli S, Verusio C, Bergamo F, Sobrero A, Frontini L, and Magnani M (2019) Sex-Related Differences in Impact on Safety of Pharmacogenetic Profile for Colon Cancer Patients Treated with FOLFOX-4 or XELOX Adjuvant Chemotherapy. *Sci Rep* 9:11527.
- Schiebinger L, Leopold SS, and Miller VM (2016) Editorial policies for sex and gender analysis. *Lancet* **388**:2841–2842.
- Schwab M, Zanger UM, Marx C, Schaeffeler E, Klein K, Dippon J, Kerb R, Blievernicht J, Fischer J, Hofmann U, Bokemeyer C, Eichelbaum M, and German 5-FU Toxicity Study Group (2008) Role of genetic and nongenetic factors for fluorouracil treatment-related severe toxicity: a prospective clinical trial by the German 5-FU Toxicity Study Group. J Clin Oncol 26:2131–2138.
- Schwartz GL, Turner ST, Chapman AB, and Boerwinkle E (2002) Interacting effects of gender and genotype on blood pressure response to hydrochlorothiazide. *Kidney International* **62**:1718–1723.
- Scripture CD, and Figg WD (2006) Drug interactions in cancer therapy. Nat Rev Cancer 6:546–558.
- Sloan JA, Loprinzi CL, Novotny PJ, Okuno S, Nair S, and Barton DL (2000) Sex differences in fluorouracilinduced stomatitis. *J Clin Oncol* **18**:412–420.
- Soldin OP, Chung SH, and Mattison DR (2011) Sex differences in drug disposition. *J Biomed Biotechnol* **2011**:187103.
- Soldin OP, and Mattison DR (2009) Sex differences in pharmacokinetics and pharmacodynamics. *Clin Pharmacokinet* **48**:143–157.
- Stein BN, Petrelli NJ, Douglass HO, Driscoll DL, Arcangeli G, and Meropol NJ (1995) Age and sex are independent predictors of 5-fluorouracil toxicity. Analysis of a large scale phase III trial. *Cancer* 75:11–17.
- Swen JJ, Huizinga TW, Gelderblom H, de Vries EGE, Assendelft WJJ, Kirchheiner J, and Guchelaar H-J (2007) Translating pharmacogenomics: challenges on the road to the clinic. *PLoS Med* **4**:e209.
- Swen JJ, Nijenhuis M, van Rhenen M, de Boer-Veger NJ, Buunk A-M, Houwink EJF, Mulder H, Rongen GA, van Schaik RHN, van der Weide J, Wilffert B, Deneer VHM, Guchelaar H-J, and Dutch Pharmacogenetics Working Group (DPWG) of the Royal Dutch Pharmacists Association (KNMP) (2018) Pharmacogenetic Information in Clinical Guidelines: The European Perspective. *Clin Pharmacol Ther* 103:795–801.
- Swen JJ, van der Wouden CH, Manson LE, Abdullah-Koolmees H, Blagec K, Blagus T, Böhringer S, Cambon-Thomsen A, Cecchin E, Cheung K-C, Deneer VH, Dupui M, Ingelman-Sundberg M, Jonsson S, Joefield-Roka C, Just KS, Karlsson MO, Konta L, Koopmann R, Kriek M, Lehr T, Mitropoulou C, Rial-Sebbag E, Rollinson V, Roncato R, Samwald M, Schaeffeler E, Skokou M, Schwab M, Steinberger D, Stingl JC, Tremmel R, Turner RM, van Rhenen MH, Dávila Fajardo CL, Dolžan V, Patrinos GP, Pirmohamed M, Sunder-Plassmann G, Toffoli G, Guchelaar H-J, and Ubiquitous Pharmacogenomics Consortium (2023) A 12-gene pharmacogenetic panel to prevent adverse drug reactions: an openlabel, multicentre, controlled, cluster-randomised crossover implementation study. *Lancet* 401:347–356.
- Swen JJ, Wilting I, de Goede AL, Grandia L, Mulder H, Touw DJ, de Boer A, Conemans JMH, Egberts TCG, Klungel OH, Koopmans R, van der Weide J, Wilffert B, Guchelaar H-J, and Deneer VHM (2008) Pharmacogenetics: from bench to byte. *Clin Pharmacol Ther* **83**:781–787.

Takayama K (2020) In Vitro and Animal Models for SARS-CoV-2 research. Trends Pharmacol Sci 41:513–517.

Tharpe N (2011) Adverse drug reactions in women's health care. J Midwifery Womens Health 56:205–213.

- Uetrecht J (2007) Idiosyncratic drug reactions: current understanding. *Annu Rev Pharmacol Toxicol* **47**:513–539.
- Van Driest SL, Shi Y, Bowton EA, Schildcrout JS, Peterson JF, Pulley J, Denny JC, and Roden DM (2014) Clinically actionable genotypes among 10,000 patients with preemptive pharmacogenomic testing. *Clin Pharmacol Ther* **95**:423–431.
- Wagner AD, Grothey A, Andre T, Dixon JG, Wolmark N, Haller DG, Allegra CJ, de Gramont A, VanCutsem E, Alberts SR, George TJ, O'Connell MJ, Twelves C, Taieb J, Saltz LB, Blanke CD, Francini E, Kerr R, Yothers G, Seitz JF, Marsoni S, Goldberg RM, and Shi Q (2021) Sex and Adverse Events of Adjuvant Chemotherapy in Colon Cancer: An Analysis of 34 640 Patients in the ACCENT Database. J Natl Cancer Inst 113:400–407.
- Wagner AD, Oertelt-Prigione S, Adjei A, Buclin T, Cristina V, Csajka C, Coukos G, Dafni U, Dotto G-P, Ducreux M, Fellay J, Haanen J, Hocquelet A, Klinge I, Lemmens V, Letsch A, Mauer M, Moehler M, Peters S, and Özdemir BC (2019) Gender medicine and oncology: report and consensus of an ESMO workshop. Ann Oncol **30**:1914–1924.
- Wang J, and Huang Y (2007) Pharmacogenomics of sex difference in chemotherapeutic toxicity. *Curr Drug Discov Technol* **4**:59–68.
- Wen L, Ley RE, Volchkov PY, Stranges PB, Avanesyan L, Stonebraker AC, Hu C, Wong FS, Szot GL, Bluestone JA, Gordon JI, and Chervonsky AV (2008) Innate immunity and intestinal microbiota in the development of Type 1 diabetes. *Nature* **455**:1109–1113.
- Wilkinson GR (2005) Drug metabolism and variability among patients in drug response. *N Engl J Med* **352**:2211–2221.
- Woodruff TK (2014) Sex, equity, and science. Proc Natl Acad Sci U S A 111:5063–5064.
- Yang L, Li Y, Hong H, Chang C-W, Guo L-W, Lyn-Cook B, Shi L, and Ning B (2012) Sex Differences in the Expression of Drug-Metabolizing and Transporter Genes in Human Liver. J Drug Metab Toxicol 3:1000119.
- Yang L, Price ET, Chang C-W, Li Y, Huang Y, Guo L-W, Guo Y, Kaput J, Shi L, and Ning B (2013) Gene expression variability in human hepatic drug metabolizing enzymes and transporters. *PLoS One* 8:e60368.
- Zalcberg J, Kerr D, Seymour L, and Palmer M (1998) Haematological and non-haematological toxicity after 5-fluorouracil and leucovorin in patients with advanced colorectal cancer is significantly associated with gender, increasing age and cycle number. Tomudex International Study Group. *Eur J Cancer* 34:1871–1875.
- Zucker I, and Prendergast BJ (2020) Sex differences in pharmacokinetics predict adverse drug reactions in women. *Biol Sex Differ* **11**:32.
- Zucker I, Prendergast BJ, and Beery AK (2022) Pervasive Neglect of Sex Differences in Biomedical Research. Cold Spring Harb Perspect Biol **14**:a039156.

# Footnotes:

This work was supported by the Italian Drug Agency (AIFA) under grant number 1729 dd. 18/11/2022 (GT).

No author has an actual or perceived conflict of interest with the contents of this article.

Table 1. List of studies addressing the interaction between germline genetic polymorphisms and patient sex in defining the risk of toxicity or inefficacy.

Drug	Number of	Gene/SNP	End-point	Main findings	Reference
	patients	investigated			
ramipril	352	ACE I/D	Efficacy	ACE inhibition is uniformly	(Ruggenenti
	(269 M; 83 F)	polymorphism	(renoprotection)	renoprotective in women regardless of the	et al., 2000)
				ACE polymorphism, and in men with the	
				DD genotype, but it is virtually devoid of	
				beneficial effects in men with the I/I or	
				I/D genotype.	
hydrochlorothiazide	376	ACE I/D	Efficacy (blood	The ACE I/D polymorphism predicts	(Schwartz et
	(170 M; 206	polymorphism	pressure control)	blood pressure response to a thiazide	al., 2002)
	F)			diuretic in a gender-specific manner: a	
				higher response rate in I/I females and	
				D/D men was observed.	
pentazocine	42	<i>MC1R</i> (R151C,	Efficacy (analgesia)	A significant influence of <i>MC1R</i> genotype	(Mogil et al.,
	hydrochlorothiazide	ramipril 352 (269 M; 83 F) hydrochlorothiazide 376 (170 M; 206 F)	ramipril 352 ACE I/D (269 M; 83 F) polymorphism hydrochlorothiazide 376 ACE I/D (170 M; 206 polymorphism F)	ramipril352ACE I/DEfficacy(269 M; 83 F)polymorphism(renoprotection)hydrochlorothiazide376ACE I/DEfficacy (blood(170 M; 206polymorphismpressure control)F)F)Image: State of the state of th	ramipril352ACE I/DEfficacyACE inhibition is uniformly(269 M; 83 F)polymorphism(renoprotection)renoprotective in women regardless of the ACE polymorphism, and in men with the DD genotype, but it is virtually devoid of beneficial effects in men with the I/I or I/D genotype.hydrochlorothiazide376ACE I/DEfficacy (bloodThe ACE I/D polymorphism predictsf)(170 M; 206polymorphismpressure control)blood pressure response to a thiazide diuretic in a gender-specific manner: a higher response rate in I/I females and D/D men was observed.

		(24 M; 18 F)	R160W, D294H		on analgesia was detected in women only.	2003)
			redhead associated		Pentazocine at the dose used produced	
			variants)		modest analgesia in all men. "Classic"	
					light-skinned, redheaded women with two	
					variant MC1R alleles displayed robust	
					pentazocine analgesia.	
2014	methadone or	582	<i>OPRD1</i> (rs1042114,	Efficacy (analgesia)	A specific sex-related effect was observed	(Clarke et
	buprenorphine/naloxone	(397 M; 185	rs678849,		for 2 intronic SNPs (rs581111 and	al., 2014)
		F)	rs10753331,		rs529520) that predicted treatment	
			rs529520, rs581111,		outcome only in females treated with	
			and rs2234918)		buprenorphine. s581111 and rs52920 may	
					be useful when considering treatment	
					options for female opioid addicts.	
2018	atorvastatin	127	ABCC2 (rs717620)	Efficacy	Triglycerides (TG) levels and the	(Prado et al.,
		(49 M, 78 F)		(triglycerides level)	TG/HDL-C ratio are affected by the	2018)
					rs717620 SNP in Chilean males but not	
					female individuals after atorvastatin	
					treatment.	

29

2020	Opioids, several molecules	172	OPRM1 variants and	Toxicity (vomiting,	Genotype of OPRM1 and COMT	(Planelles et
		(45 M; 127 F)	COMT	depression,	(OPRM1-G allele and COMT-GG	al., 2020)
				dizziness, sexual	polymorphisms) influenced toxicities for	
				dysfunction)	vomiting and depression in men, dizziness	
					in women and sexual dysfunction in both.	
2021	omeprazole, esomeprazole,	468.280	<i>CYP2C19</i>	Toxicity (migraine)	Treatment with proton-pump inhibitors	(Pisanu et
	lansoprazole, pantoprazole	(214.285 M;			and CYP2C19 poor metabolizer status	al., 2021)
	and rabeprazole	253.995 F)			were associated with higher incidence of	
					probable chronic migraine exclusively in	
					men	
2022	oxycodone/naloxone	584	OPRM1 (rs1799971,	Toxicity (erythema	Patients with COMT-472AA homozygote	(Barrachina
	tapentadol	(169 M; 415	A118G) and COMT	and vomiting)	genotype evidenced higher rates of	<i>et al.</i> , 2022)
		F)	(rs4680, G472A)		erythema and vomiting, especially in	
					females	

SNP, Single Nucleotide Polymorphism; M, Males; F, Females

Table 2. List of studies addressing the interaction between germline genetic polymorphisms and patient sex in defining the risk of fluoropyrimidines-related toxicity.

Year	Drug	Number of	Gene/SNP investigated Main findings		Reference
		patients			
2008	5-FU	683	DPYD, TYMS, MTHFR	A significant interaction was found between sex and	(Schwab <i>et al.</i> , 2008)
		(383 M; 300 F)		DPYD*2A, which resulted much more associated with	
				toxicity in males than females	
2014	5-FU	2594 (1385 M;	DPYD	A greater effect of <i>DPYD</i> *2A (associated with grade $\geq 3$	(Lee et al., 2014)
		1209 F)		5FU-toxicities) in males compared with females.	
2019	5-FU	512	<i>TYMS</i> (rs34743033,	Interactions were detected on time to grade $\geq 3$	(Ruzzo <i>et al.</i> , 2019)
		(294 M; 218 F)	rs2853542, rs11280056),	haematological toxicity (TTH) for rs1801133 and	
			MTHFR (rs1801133,	rs1799793, on time to grade $\geq$ 3 gastrointestinal toxicity	
			rs1801131), ERCC1	(TTG) for rs13181, and on time to grade $\geq 2$ neurological	
			(rs11615), XRCC1 (rs25487),	toxicity (TTN) for rs11615.	
			XRCC3 (rs861539), XPD	Rs1799793 GA genotype and A allele shortened TTH in	
			(rs1799793, rs13181), GSTP1	men. In women, the rs11615 CC genotype worsened	
			(rs1695), GSTT1/GSTM1	TTN and rs13181 G allele improved the TTG.	
			(deletion +/-), <i>ABCC1</i>		

			(rs2074087), and <i>ABCC2</i>		
			(rs3740066, rs1885301,		
			rs4148386).		
2021	5-FU/	313	<i>TYMS</i> -TSER (rs45445694)	TYMS-TSER 3R/2R polymorphism was associated with	(Ioannou <i>et al.</i> ,
	capecitabine	(160 M; 153 F)	polymorphism	incidence of adverse events in female cancer patients.	2021)
2021	5-FU/	126	<i>TYMS</i> -TSER (rs45445694)	A higher prevalence of 2R/2R TYMS genotype was	(Khushman et al.,
	capecitabine	(69 M; 57 F)	polymorphism	reported in female African American patients developing	2021)
				adverse events compared with men.	
2022	5-FU/	313	<i>MTHFR</i> rs1801133 C>T	MTHFR rs1801133 polymorphism was associated with	(Ioannou <i>et al</i> .,
	capecitabine	(160 M; 153 F)	polymorphism	fluoropyrimidine dose reduction in female cancer	2022)
				patients.	

SNP, Single Nucleotide Polymorphism; M, Males; F, Females; 5-FU, 5-Fluorouracil; R, repeat

# Figure 1

